Vol. 5 **NO** 2 2017 - Novembro

Revista de Defesa da Concorrência PUBLICAÇÃO OFICIAL



2318 2253

Data-driven mergers: a call for further integration of dynamics effects into competition analysis

Andressa Lin Fidelis³⁹⁰

ABSTRACT

This report reviews data-driven mergers by focusing on two main competition challenges. Firstly, it analyses whether data-platforms may pose an unreplicable competitive advantage to entrants, and to what extend data-driven indirect network effects may harm competition and innovation. Secondly, it reviews the intersection between competition and privacy, and the quantification of the effects of quality reduction. The report argues that a more dynamic approach can contribute to address those challenges better than a pure static framework, and proposes a re-analysis of the European Commission's merger decision regarding Facebook/WhatsApp. The report concludes by proposing that dynamic effects, consumer choice, and merger control be analysed more holistically.

Keywords: Facebook/WhatsApp, competition economics, merger control, big data, dynamic analysis, network effect, privacy

RESUMO

O presente artigo explora dois desafios ao direito concorrencial decorrentes dos atos de concentração envolvendo *big data*. Primeiramente, analisa-se em que condições as plataformas digitais podem representar uma vantagem competitiva não replicável e como as externalidades indiretas decorrentes do *big data* poderiam prejudicar a concorrência e a inovação no mercado. Em segundo lugar, foca-se na intersecção entre concorrência e privacidade, sugerindo análises quantitativas para sua mensuração. Este trabalho argumenta que a adoção de um perspectiva mais dinâmica pode endereçar ambos os desafios de modo superior à análise puramente estática, propondo a revisão da aprovação do ato de concentração Facebook/WhatsApp pela Comissão Europeia. O trabalho conclui propondo que os efeitos dinâmicos, a escolha do consumidor e o controle de estruturas sejam analisados de forma mais holística.

Palavras-chave: Facebook/WhatsApp, economia da concorrência, atos de concentração, big data, análise dinâmica, externalidades de rede.

³⁹⁰ Andressa Lin Fidelis holds an LL.M. with focus on US and EU competition law from Georgetown Law (2014) and a Master in Applied Economics from Barcelona Graduate School of Economics (2017). Previously, she worked as an associate in a law firm in Sao Paulo, as a trainee at the US Federal Trade Commission and served as a Coordinator at CADE's Leniency Unit. Ms. Fidelis currently works in the competition team at Latham & Watkins, Brussels' office. The author thanks Zeynep Ortac (BGSE 2017) for the research and first draft on the intersection between data and privacy. For further contact: <u>alf62@georgetown.edu</u>

SUMÁRIO: 1. Introduction; 2. Data-driven markets: can concentrated market still be competitive?; 2.1. Setting the debate: concentration vs. market power; 2.2. Data as a competitive advantage; 2.3. Entry barriers & data-driven indirect network effects; 2.4. Market tipping & innovation; 3. Non-price dimension of competition: does privacy matter?; 3.1. Privacy as a dimension of quality competition; 3.2. The economics of privacy: 3.3. Ouantitative analysis of effects on quality; 4. Incorporating a more dynamic approach into merger analysis: what does it mean?; 4.1. Forwardlooking approach to market definition; 4.2. Connected markets and the domino effect; 4.3. Do we need new tools?; 5. Could a dynamic analysis have changed the outcome of Facebook/WhatsApp?; 5.1. A dynamic market for data; 5.2. Elimination of potential competition; 5.3. Tipping a connected market; 5.4. Privacy and quality degradation; 6. Conclusion; 7. References.

1. Introduction

Data-driven mergers are the transactions that aim at acquiring, combining and/or monetizing large amounts of commercially valuable data gathered from multiple sources and formats³⁹¹. In the digital markets (e.g., e-commerce, social networks, search engines, online advertisement, etc.), examples can be found in the mergers between *Verizon/Yahoo!* (2016), *Microsoft/LinkedIn* (2016), *Facebook/WhatsApp* (2014), *Google/DoubleClick* (2008)³⁹², among many others³⁹³.

Those transactions benefit from the developments of artificial intelligence, data mining and machine learning, allowing data to be analyzed for insights that can reduce product and process innovation costs. Indeed, consumers' data is at the core of the business model³⁹⁴ and

³⁹¹ Doug Laney crafted the pioneer definition of big data in three dimensions: (i) volume: data comes in large amount and it is collected from a variety of sources such as business transactions, social media, information from sensor, machine-to-machine, etc.; (ii) velocity: data streams in at an unprecedented speed and must be dealt with in near-real time; and (iii) variety: data can be structured and unstructured and comes in all types of formats, e.g., numeric or text documents, e-mail, video, audio, etc.. See Laney (2001). More recently, extra dimensions have been added, including: (iv) variability; (v) veracity; (vi) validity; (vii) vulnerability; (viii) volatility; (ix) visualization; and (x) value. See Firican (2017).

³⁹²Verizon/Yahoo!, case COMP/M.8180, EC's Decision on 21.12.2016; *Microsoft/LinkedIn*, case COMP/M.8124, EC's Decision on 6.12.2016; *Facebook/WhatsApp*, case COMP M.7217, EC's Decision on 03.10.2014, and; *Google/DoubleClick*, case COMP/M.4731, EC's Decision on 22.07.2008. Other big data mergers include *Telefónica UK/ Vodafone UK/ Everything Everywhere/JV*, case COMP/M.6314, EC's decision on 04.09/2012 and *Publicis/Omnicom*, case COMP/M.7023, EC's decision on 09.01.2014

³⁹³ As reported by The Economist (May, 2017), recent data-driven deals also involved: Facebook/Instagram for \$1 bn (2012), Alphabet/Waze for \$1.2 bn (2013), IBM/The Weather Company for \$2 bn (2015), IBM/Truven Health Analytics for \$2.6 bn (2016), Intel/Mobileye for \$15.3 bn (2017), Microsoft/SwiftKey for \$0.25 bn (2016), Oracle/BlueKai for \$0.4 bn (2014), Oracle/Datalogix for \$1bn (2014).

³⁹⁴Described as a "raw material for digital business models", personal information has become a factor of competition used to improve products and targeted advertising. See Monopolkomission (2015, p. 36).

largely explain the market power enjoyed by the world's most valuable public companies, namely Apple, Alphabet, Microsoft, Amazon, and Facebook³⁹⁵.

This reports focus on two main competition challenges posed by data-driven mergers. First, the trend of increasing concentration of super-platforms is prompting a debate over whether or not we need new tools and more severe competition enforcement to guarantee contestability in data platforms³⁹⁶. Is market power entrenched or entry hampered when bigger data firms take over smaller ones?

The second challenge is related to the intersection between competition and privacy. Data-driven platforms generally offer services for no monetary fee to one side of the platform. When services are offered for "free", how authorities can measure the effects of a merger in non-price dimension of competition, such as quality and privacy?

This report argues that a more dynamic analysis can contribute to address those challenges better than a pure static competition analysis. Where predictive fact-finding can be supported by economic theory and empirical evidence, a dynamic approach is more suitable to address competition concerns in rapidly evolving data-driven economies. Therefore, a re-analysis of the European Commission ("EC") approval decision regarding Facebook/WhatsApp is suggested to test whether its outcome would have changed if a more forward-looking analysis were taken into account.

The remaining of this report is organized as follow: Section 2 analyzes the main economic characteristics of data-driven markets, and how user information can yield those platforms an unreplicable competitive advantage. Section 3 discusses introducing privacy considerations into the competitive assessment. Section 4 explains how dynamic analysis can contribute to improve data-driven mergers' review. Section 5 evaluates what could have changed if a more dynamic analysis were adopted in the Facebook/WhatsApp merger analysis. Section 6 concludes by summarizing the main takeaways and proposing that dynamic effects, consumer choice, and merger control be analyzed more holistically.

2. Data-driven markets: can concentrated market still be competitive?

³⁹⁵ Based on the Financial Times Global 500 ranking of 2017, Apple, Alphabet (Google), Microsoft, Amazon, and Facebook are among the 8 publicly traded companies having the greatest market capitalization.

³⁹⁶ For those who are promore enforcement, see, e.g., the "Brandeisian Movement" summarized by Dayen (2017), The Economist (May 2017 and September 2016), OECD (2016, p. 20-24), EDPS (2016, p. 7-13), Khan (2017), Taplin (2017), Ezrachi and Stucke (2016), Thompson (2016), Dayen (2016), and Lanier (2014). For those who are against, see e.g., Lamadrid and Villiers (2017), as well as Kevin Murphy at the Conference: *Does America have a concentration problem?* March 28, 2017, Chicago, US.

Data-driven markets often give consumers the impression that competition is fiercer, prices are more transparent³⁹⁷, and monopolization is less likely. Some claim that data-driven markets have nearly zero entry barriers³⁹⁸ because data is ubiquitous, low cost, and widely available. Consequently, market shares are clearly not a good proxy of market power because competition is just "one click away" (Schmidt, 2015). At anytime, coming from a garage somewhere, Schumpeterian "gales of creative destruction" could displace Google or Facebook in the same way that they displaced Yahoo! and MySpace. Thus, most calls for more antitrust intervention against data-driven platforms should be regarded as merely "antitrust populism" (Lamadrid and Villiers, 2017, p. 3).

On the other hand, many consider that data's nature favors long lasting dominance and are concerned about the entrenchment of data-driven platforms to the detriment of innovation³⁹⁹, consumer welfare⁴⁰⁰, and even democracy⁴⁰¹. To navigate in this debate, this report focus on three main characteristics of data-driven markets: (i) concentration as a proxy for market power; (ii) data as an irreplicable asset; (iii) higher entry barriers mainly caused by indirect network effects; and (iv) the tendency for market tipping in favor of the data-driven incumbent.

2.1 Setting the debate: concentration vs. market power

Apple, Alphabet/Google, Microsoft, Amazon and Facebook have enjoyed high market shares, reaching billion of users, and significant profits⁴⁰², some for over a decade now (EDPS, 2016). Data-driven markets have an intrinsic trend of becoming even more concentrated (OECD, 2016). Such digital concentration has prompted calls for the tech giants to be broken up (as reported by The Economist, 2017). The so-called Brandeisian movement considers that

³⁹⁷ On the debate about how price transparency can backfire via anticompetitive price discrimination and logarithm collusion, see OECD (2016), and The Economist (May 6, 2017).

³⁹⁸ In November 26, 2012, Ron Wyden, U.S. senator, declared in the FTC's investigation against Google's search bias case: "*compared to almost any other market in the history of antitrust regulation, online search has effectively zero barriers to entry*", https://www.wyden.senate.gov/download/?id=94C57310-59D3-4D6E-84BE-FF957413BCC3&download=1.

³⁹⁹ As observed by The Economist (September 17, 2016) the number of startups in the US is the lower since the 1970s.

⁴⁰⁰ Authors like Stucke and Grunes (2017 and 2016) have tackled the commonly asserted "myths" regarding the contestability of data-driven markets.

⁴⁰¹ See Dayen (2017).

⁴⁰² According with Jonathan Taplin, Alphabet/Google has 88% search advertising and is valued at \$530.6 bi, Amazon has 75% in online book sales and is valued at \$362.4 bi, and Facebook/Instagram/WhatsApp 74% market share of mobile social and is valued at \$357 bi. Google and Facebook have each nearly 2 billion customers. See the Conference: *Does America have a concentration problem?* March 28, 2017, Chicago, US, https://research.chicagobooth.edu/stigler/events/single-events/march-27-2017.

the fear of antitrust agencies of overreaching fails to serve the public⁴⁰³. In face of the "tsunami" of digital mergers, US Senators doubt that the agencies are getting the job done (Dayen, 2016). The other side of the spectrum criticizes the emphasis on concentration as misplaced. For Sidak and Teece (2009, p. 36-37), policymakers should even overturn the presumption that more competitors are always better, as "high market share may indicate not only superior performance, but also strong selection at work in the industry".

While competition law is not merely concerned with concentration, but with market power and the ability of dominant firms to subvert the competitive process (Hesse, 2016), I believe that concentration levels are an important screening device for competition authorities. Data innovation has brought high R&D level, and benefit consumers with new products at even zero monetary fees. However, as seen below, data characteristics can help perpetrate the quite concentrated nature of data-driven platforms (Graef, 2015). Thus, the remaining question is: can the informational power become monopoly power?

2.2 Data as a competitive advantage

Due to the alleged non-rivalrous⁴⁰⁴ nature of data, some authors claim that no incontestable market power could be derived from it (Tucker and Wellford, 2014). Indeed, factual information such as name, age, gender, home address, etc. are commonly provided to multiple entities, but they are not the kind of inputs that search engines, social networks or e-commerce need to provide relevant services to both sides of their platforms (CMA, 2015, paras 2.53-2.54). The volume, scope, and precision of analysis of data gathered nowadays cannot be compared to a brick-and-mortar world.

Successful data-driven platforms have an established user base allowing them to collect, store and process large, real-time data about users last minute interests, sentiments, influence, and behavior⁴⁰⁵. Specifically, search engines care about search queries history and clicked links. Social networks care about profile information, and constructing the user's social graph. E-

⁴⁰³ As described by Dayen (2017, p. 4-5), the "New Brandeis movement" is formed by a group of scholars that has rebelled against Chicago-school dictates. For the Supreme Court Judge Louis Brandeis: "we can have democracy in this country, or we can have great wealth concentrated in the hands of a few, but we can't have both."

⁴⁰⁴ A non-rivalrous good means that the cost of providing it to a marginal individual is zero (Cornes and Sandler, 1968). In big data, it means that the same data may be used by different firms at the same time.

⁴⁰⁵ This report acknowledges that big data has to be processed by logarithms and AI to become valuable. But the comparative importance of the former compared to the later has been addressed even by Google's chief scientist in 2010, who peremptorily affirmed that "we don't have better algorithms than anyone else. We just have more data". See Asay and O'Reilly (2010).

commerce cares about users' purchase history. This information is far from being easily collected and it is not readily available on the market (Grunes and Stucke, 2015).

Moreover, the fixed costs involved in setting up the necessary tools for collecting and analyzing data are high. Third party access is also a remote possibility. While platforms like Facebook and LinkedIn prohibit third parties from scrapping content off its platforms, Google restricts portability of advertising campaigns (Graef, 2015), not to mention all the patents involved. The reason for this protection wall is that for data-driven markets, data is the input that strength both sides of their platforms: (i) advertisers benefit from better targeted advertising business possibilities; and (ii) users benefit from the higher quality of the functionalities offered.

Big data advantage also allows dominant platforms to closely oversight (or nowcast⁴⁰⁶) not only consumers' behavior and markets' trends, but also the development of rivals' business model and nascent threats. No wonder why almost every (realistic) start-up's dream is to be acquired by, rather than become the next big data titan (The Economist, September 2016).

Regardless of being considered, the "new currency of the internet" (Vestager, 2016), the oil of the XXI century (The Economist, May 2017), the important antitrust factor is that big data is commercially valuable – as showed by the value of their billionaire acquisitions⁴⁰⁷ –, a driver of change, a competitive advantage for incumbents, and entry barrier for entrants (Graef, 2015). Depending on other characteristics of the data-driven market (e.g., market foreclosure, replicability, double marginalization problem, etc.), the essential facility doctrine may also apply⁴⁰⁸.

2.3 Entry barriers & data-driven indirect network effects

⁴⁰⁶ Nowcasting is defined as the prediction of the present, the very near future and the very recent past in economics, https://en.wikipedia.org/wiki/Nowcasting_(economics).

⁴⁰⁷ Only *Facebook/WhatsApp* and *Microsoft/LinkedIn* transactions amount for more than \$48 billions.

⁴⁰⁸ In the EU, the legal test and threshold for treating an input as an essential facility is significantly high, see the ECJ in *IMS Health* (case C- 418/01, para. 34-52) and *Bronner* (*case*, C-7/97, para.44-45). The US case law is even more averse to applying the doctrine. From an economic point of view, the author believes that a dynamic framework may be more suitable to address data issues than the essential facility doctrine.

With respect to entry barriers, the challenge pose by data-driven markets is that on top of the traditional network effects⁴⁰⁹, there are the effects derived from scale⁴¹⁰ (i.e., learning-by-doing) and scope⁴¹¹ (i.e., multiple data aspects of one user), and what some author have called spill-over or "data-driven indirect network effect" (Prüfer and Schottmüller, 2017), which have widen the gap between incumbent and entrants relevance of service. Indeed, the existence of an indirect network effect that crosses customer groups is what characterizes a business as multi-sided⁴¹² (Graef, 2015, p. 476).

Prüfer and Schottmüller (2017) propose an innovative dynamic model of R&D competition to show that, due to indirect network effects, data-driven markets become stably monopolized ("tip") under very mild conditions⁴¹³. The model's fundamental mechanism is to treat demand side-generated user preferences or characteristics as an input into the supply side-run innovation process. Thus, user information is an input into a firm's efforts to improve its perceived product quality and therefore reduces firm's cost of innovation. In the case of search engines, e.g., users' clicking behavior is the driver for indirect network effects.

As explained by the authors, a key feature of the "datafication" process is the growing importance of the indirect network effect, which combine the machine-generated data about user information, as a by-product of using goods and services that are connected to the internet, with a reduction in the marginal cost of innovation on the supply side, s.t. the marginal cost of innovating $c(x, D_i)$, is decreasing in demand: $c_{x, Di} < 0^{414}$.

As concluded by the authors, such combination cannot be easily copied by rivals or overcome by a disruptive innovator. Thus, data-driven markets do seem to pose higher barriers to entry resulting from the indirect network effects. Indeed, as observed by Stucke and Grunes

⁴⁰⁹ A network effect is either direct when the consumers' utility increases as the number of consumers grows, or indirect when the increasing number of consumers of a good leads to more complementary products or services that raises the value of the network (Katz and Shapiro, 1985). In the case of search engine services, the indirect network externality lies on the fact that the search results increase in relevance the more search data become available to the search engine (Argenton & Prüfer, 2012).

⁴¹⁰ Economies of scale arise when the incremental costs of creating additional units decline as the scale of production increases (Shapiro and Varian, 1999). In data-driven markets, users and advertisers expect to gain more value and are attracted to platforms with the largest group of customers.

⁴¹¹ As suggested by Stucke and Grunes (2017), entry barriers are originated from four network effects: (i) classic network effects; (ii) network effects arising from the scale of data; (iii) network effects from the scope of data; and (iv) how network effects on one side of a platform can spill over to the other side.

⁴¹² I agree with Prüfer and Schottmüller (2017) who does not define search engines and social networks as twosided, but as *semi*-two-sided, as a higher number of advertisers do not necessarily benefit users.

⁴¹³ The authors adopt a model where duopolists repeatedly choose their innovation investments to compute the subgame-perfect Nash equilibria with a finite time horizon.

⁴¹⁴ According with the model, the firms have to incur in the following investment cost to increase its quality by x unit: $c(x,Di) = \gamma x^2/2 + \alpha x [1-Di(\Delta)]$, where γ is a parameter that measures the difficulty to innovate, α is a parameter that measures data-driven indirect network effects, Di is the demand in T-1, and Δ is the quality difference between the 2 firms.

(2017), if barriers to entry were low, Google wouldn't have intentionally degraded quality to favor its vertical service, and Microsoft wouldn't have spent over \$4.5 billion to develop the algorithm and capacity to operate Bing – without virtually any success.

2.4 Market tipping & innovation

Prüfer and Schottmüller (2017) show that there is a strong first-mover advantage in datadriven markets, which tip under very mild conditions. In the game proposed by the authors, it means that when quality difference is sufficiently large, firms do not value future too much (δ is sufficiently low), and innovating is not so expensive, eventually, one firm will dominate the market by having full demand in every second period.

An alarming feature of a tipped market is that "there are very little incentives for both the dominant firm and the ousted firm to further invest in innovation" Prüfer and Schottmüller (2017, p. 2). This is because, by backward induction, the smaller firm will choose not to invest in innovation since it knows that the dominant firm will be able to match any investment at a lower marginal cost. Knowing this, the dominant's best response is also not to invest. Thus, market tipping cannot only raise barriers to entry, but it can also harm consumers due to the resulting underprovision of innovation.

When a market favors a "winner takes all outcome where monopoly is the nearly inevitable outcome of market success" (OECD, 2014, p. 60), competition authorities should have more reasons to be concerned about mergers that can tip the market, either in the service or product market where the acquirer already enjoys a large share, either in connected markets in which the acquirer can leverage its position, using information gained in his market of origin to tip a second or multiple markets (domino effect) (Prüfer and Schottmüller, 2017), as detailed in Section 4 below.

Alternatives to guarantee market contestability in data-driven markets include fostering consumers' data ownership and portability initiatives⁴¹⁵. Also, data sharing of anonymized user information could allow competitors to overcome the incumbent's data advantage (The

⁴¹⁵ In the EU, beginning in May 2018, the Regulation 2016/679 will impose an obligation on firms to enable individuals to take their personal data with them when they quit using an online service.

Economist, March⁴¹⁶ and May⁴¹⁷, 2017), while eliminating the mechanism causing data-driven markets to tip⁴¹⁸⁻⁴¹⁹.

In sum, data can become an insuperable competitive advantage when: (i) acquirer's dominant position has been stable for a significant period; (ii) data is not easily replicable and the incumbent relies on exclusivity and IP rights; (iii) data-driven indirect network effects are strong; and (iv) the mergers can help the product market or a correlated market to tip.

3. Non-price dimension of competition: does privacy matter?

A survey published by International Data Corporation in 2017 found that "*a whopping* of 84% of U.S. consumers expressed concern regarding the security of their personally identifiable information and 70% told IDC that their concern is greater today than just a few years ago"⁴²⁰. In response to the scaling concern about privacy protection, privacy-conscious search engines such as DuckDuckGo and Qwant have emerged offering a differentiated product. As sharing private information with data-driven platforms is increasingly being perceived by consumers as a "cost", it is questionable whether the services offered by search engines and social networks can be considered cost free (OECD, 2016). In fact, data-driven mergers may diminish the incentives of the merging parties to compete on quality dimensions such as privacy protection, and, therefore, substantially hinder competition.

However, privacy concerns are far from being incorporated in competition analysis, with the notorious exception of Bundeskartellamt's investigation on Facebook's alleged abuse of privacy terms⁴²¹, and the Italian Competition Authority investigation on WhatsApp's Term of Use⁴²². In the merger realm, both the EC and the FTC have cleared the Google/DoubleClick

⁴¹⁶ In the EU, beginning in January 2018, the Second Payment Service Directive (PSD2) will compel banks to share customer-account information with licensed financial-services providers, under the consent of the account-holder.

⁴¹⁷ Ben Thompson, suggested that dominant social networks should be required to allow access to their social graphs, and highlighted that Instagram got off the ground by having new users import the list of their followers from Twitter.

⁴¹⁸ If, e.g., Facebook does not have exclusive rights of user information, competitors face the same cost function, and there is no cost advantage in producing quality. See Prüfer and Schottmüller (2017).

⁴¹⁹ Effects on total welfare are mixed because if there is no tipping, investments costs are duplicated. However, if data-driven indirect network effects are sufficiently high, data sharing obligations can increase total welfare. See Prüfer and Schottmüller (2017).

⁴²⁰IDC Press Release, January 24, 2017, <u>http://www.idc.com/getdoc.jsp?containerId=prUS42253017</u>.

⁴²¹ The Bundeskartellamt framed Facebook's Terms of Service on the use of consumer data as an abused of its dominant position in the market for social networks. See Bundeskartellamt's Press Release on March 2, 2016, http://www.bundeskartellamt.de/SharedDocs/Meldung/EN/Pressemitteilungen/2016/02_03_2016_Facebook.htm

⁴²² On May 11th, 2017, the ICA closed its two investigations initiated in 2016 and fined WhatsApp for \in 3 million for having forced its users to share personal data with Facebook. The ICA concluded that WhatsApp *de facto* forced its users to share their personal data with Facebook, by inducing them to believe that they would continue

merger without accounting for the privacy aspects, with the exception of a dissenting statement from former FTC Commissioner Pamela Jones Harbour where she considered privacy concerns as 'cognizable'⁴²³. The EC in Facebook/WhatsApp dismissed privacy concerns as outside the scope of EU competition law and the ECJ *Asnef-Equifax*⁴²⁴ found that privacy concerns are not, "as such", a matter for competition law.

To the extent that privacy constitutes a dimension of competition affecting economic efficiency and long run consumer welfare, competition law enforcement should step in. Having observed that privacy concerns have remained as a neglected matter in merger control, this report analyzes privacy from a quality competition dimension, investigates the competition concerns, and the importance of incorporating them into merger analysis.

3.1 Privacy as a dimension of quality competition

Quality is considered to be the key non-price competition parameter and the driver of innovation and dynamic efficiency (Gal and Rubinfeld, 2016). The European Court of Justice has long ago confirmed quality as another competition parameter⁴²⁵. As recognized by OECD (2013), lower quality can be just as detrimental to consumer welfare as higher prices. As recognized by the EC, when services are offered for no monetary fees, "*consumers pay more attention to other features*" and therefore quality constitutes "*a significant parameter of competition*"⁴²⁶. In *Microsoft/LinkedIn* the EC acknowledged the relevance of privacy in the competitive assessment⁴²⁷.

to use the service only if they grant their full consent⁴²². See ICA's press release on Press Release, <u>http://www.agcm.it/en/newsroom/press-releases/2380-whatsapp-fined-for-3-million-euro-for-having-forced-its-users-to-share-their-personal-data-with-facebook.html</u>

⁴²³ "[W]ithout imposing any conditions on the merger, neither the competition nor the privacy interests of consumers will have been adequately addressed". See Pamela Jones Harbour, Dissenting Statement, In the Matter of Google/DoubleClick, FTC File No. 071-0170 (2007).

⁴²⁴ Case C-238/05, ECLI:EU:C:2006:734, *Asnef-Equifax v. Ausbanc*, para 63.

⁴²⁵ "Competition on the merits may, by definition, lead to the departure from the market or the marginalization of competitors that are less efficient and so less attractive to consumers from the point of view of, among other things, price, choice, quality or innovation". Judgment of the ECJ of March 27, 2010 in Case C-209/10 Post Danmark A/S v. Konkurrencerådet, para 22.

⁴²⁶ Case M COMP/M.6281, *Microsoft/Skype*, Decision on 07.10.2011, para 81.

⁴²⁷ "Privacy (...) can be taken into account in the competition assessment to the extent that consumers see it as a significant factor of quality, and the merging parties compete with each other on this factor. In this instance, the Commission concluded that data privacy was an important parameter of competition (...) ". Press Release of December 6, 2016, http://europa.eu/rapid/press-release_IP-16-4284_en.htm.

However, OECD (2013) identifies the question of quality effects as one of the most troublesome issue in merger control context, as it is "subjective" and hard to measure⁴²⁸. E Even if competition authorities account for quality competition, the scope of such analysis remains limited to its link with the price or competition in the market (Evans, 2011)⁴²⁹. Hence, the traditional analysis and economic tools are not likely to capture the anticompetitive effects brought by mergers between firms differentiated by privacy protection levels.

3.2 The economics of privacy

As quality and privacy competition has a common interpretation, economics of privacy sheds lights to additional themes and challenges associated with the competition analysis of privacy.

First of all, there is information asymmetry between the users and the service providers. When subscribing to online platforms, consumers are rarely aware of the type of data that might be collected form them (behavioral data, IP and location tracking data, business transaction, etc.) (OECD, 2106). This asymmetry enables the platform to invest less on privacy and increase its expected profits to the expense of service quality. Even if there are data protection standards and policies, the asymmetry of information does not change since it is costly for the user to read and understand the lengthy and complex terms and conditions⁴³⁰.

This situation leads to a dysfunctional equilibrium characterized by Farrell (2012) as "few consumers devote attention to disclosures, disclosures are vague, noncommittal, or even if explicit, mostly ignored; and the privacy policies chosen are inefficiently non-protective"⁴³¹. In such equilibrium, firm will be only interested in the privacy policy that gives more follow-on revenues, without considering what the consumers would prefer.

Secondly, not just naive users but also sophisticated consumers might not be able to switch to another service provider, either because they lack bargaining power or a credible

⁴²⁸ OECD Report (2013) expresses the difficulty relating to introducing quality considerations into competitive assessment: "First, quality is a subjective concept and therefore much harder to define and measure than prices. In addition, microeconomic theory offers little help in predicting how changes in the level of competition in a market will affect quality and it is usually up to empirical analysis to determine how quality will change in response to varying degrees of competition in the context of particular markets"

⁴²⁹ A similar observation is also made by Ezrachi, A. and Stucke, M. E. (2015). Furthermore, a delegate's statement of "the existing SSNIP test already incorporates a quality assessment, because customers will take into account the quality attributes of potential substitutes when deciding whether to switch in response to a price rise" included in the OECD (2013) Report illustrates the tendency to link quality with price.

⁴³⁰ Stucke and Grunes (2016) estimates the opportunity cost involved in reading privacy policies and general terms and condition at around 10 days a year.

⁴³¹ Importantly, the author also expresses that a small firm cannot break the equilibrium by adopting more protective policies as its demand would not shift as much to worth sacrificing.

outside option⁴³². In fact, the results of the survey conducted Pew Research Center indicates that "91% of the adults 'agree' or 'strongly agree' that consumers have lost control over how personal information is collected and used by companies. By extension, the fact that the online platforms' services are presented as "free" leads consumers to disregard or at least misperceive the real cost of sharing their data. This has been also highlighted by the EC Commissioner Margrethe Vestager as "(*v*)*ery few people realize that, if you tick the box, your information can be exchanged with others (...) You give away something that was valuable. I think that point is underestimated as a factor as to how competition works"⁴³³. In particular, the user immediately benefits the "zero-price" service yet remains uninformed regarding the short or long-term consequences of the disclosing its private data (OECD, 2016).*

For instance, the economics literature suggests that effects of targeting are not necessarily always positive for consumers since they might be offered products inferior to the ones they would have found otherwise (Acquisti et al., 2016). Evans (2011) observes that zero is just another number that raises the red flag that conventional analysis do not apply for product in question and a proper antitrust analysis should consider the free product together with its companion moneymaking product.

When merger analysis involves a free product in multi-sided market, is widely known that the analysis needs to account for both sides of the platform since any change in market conditions for customers of one product affects the customers of the other product (Evans, 2011). However, as well observed by the *Autoritat Catalana de la Competència* (2016, p. 8), if there is a tradeoff between loss of privacy on the one side, and a more personalized service on the other side, the final decision should be given by the consumer.

Thirdly, when it comes to privacy, the typical assumption that consumers could detect the degradation in quality and would want to switch to rival products or services is highly unlikely to hold true (Ezrachi and Stucke, 2015). Therefore, the competition authorities should aim at incorporating the post-merger effects on parameters such as privacy into the balancing equation, albeit it may remain solely on qualitative evidences⁴³⁴. The challenge regarding

⁴³² OECD Report (2016), para 94 refers to Hoofnagle and Whittington's (2014) proposal of granting ownership rights for consumers over their in order increase their bargaining power to negotiate the conditions under which data is traded and even to receive a monetary compensation. On the other hand, Acquisti et al. (2016, p. 453), argues that while the assignment of property rights is generally welfare enhancing, granting consumers the right to sell their personal data may actually undermine consumer surplus.

⁴³³ Mlex interview with Margrethe Vestager, 22 January 2015.

⁴³⁴ The OECD Report (2013, p. 161) on the difficulties of obtaining sufficient data to conduct quantitative analysis of quality aspects: "*This is due, in large part, to the multidimensional and subjective nature of product quality itself. Accordingly, DG Competition tends to rely more heavily on qualitative evidence of quality, including*

assessing the qualitative factors could be overcome considerably through cooperating with the data protection authorities' experts on the review of the case.

3.3 Quantitative analysis of effects on quality

Lastly, there are some promising steps taken towards quantifying effects on quality dimension as well. OECD (2013) presented SSNDQ test to measure the effect of a 'small but significant non-transitory decrease in quality' within market definition context⁴³⁵. The idea behind this test is similarly to the SNNIP (small but significant, non-transitory increase in price) test, and it allows for discovering if an existing supplier were to degrade quality a certain extent⁴³⁶, holding other conditions unchanged, whether substitution to other products/services occurs. Hence, despite quantifying quality is a more challenging exercise than price, consumers' conduct may still provide rough indications about their preferences, when quality changes⁴³⁷.

Although SNNDQ test found some positive reactions and was even applied by few authorities, a disagreement regarding its applicability or necessity was evident. In response to these skepticisms, (Waehner, 2016) introduced a framework suggesting that concerns stemming from "free service" characteristic do not need to hamper the quantification of non-price effects of a merger. Analogous to the upward pricing pressure ("UPP") formula, Waehner (2016) derived the downward quality pressure ("DQP") associated with a merger by quantifying the effects on quality but without quantifying quality or privacy⁴³⁸.

The DQP formula assumes that through merger, a merged entity can internalize some of the lost profits by recapturing the users that would have otherwise switched to a competitor in response to lower privacy protection and hence, profitably and unilaterally degrade privacy⁴³⁹. Accordingly, when the price is constrained, a merger between two online advertising

evidence gathered during sector investigations, customer and competitor surveys and internal documents of the firm(s) under investigation".

⁴³⁵ This test, while sometimes applied in industries where quality measures are well-accepted and quantifiable (e.g., health sector), is used sparingly in other industries, where appropriate measures of quality have still to be developed (OECD, 2013).

⁴³⁶ Hartman et al. (1993, p. 340) argue that SSNDQ test would lead to more accurate results in markets that are subject to rapid technological development. The authors also suggested a 25% decrease in a major performance (quality) compared to the 5-10 percent increase in SSNIP test.

⁴³⁷ Gal and Rubinfeld (2016), refers to SSNIC test suggested by Newman where consumers pays for the free good in terms of another currency such as attention or information, such costs (e.g. privacy costs) can be quantified in terms of changes in cost.

⁴³⁸ Waehrer (2016, p. 20-21) asserts that it is possible to calculate a critical level of incremental cost efficiencies from a merger such that efficiencies below that level would create downward pressure on quality. He also argues that premerger margins are the same inputs for both UPP and DQP and diversion rations associated with price effects need not be that different from diversion ratios associated with SSDQ.

⁴³⁹ Just as with price effect, the value of the recaptured sales will be the quantity of diverted users times the per unit profit margin on those users.

supported firms may lead to reduction in consumer welfare if certain conditions are satisfied⁴⁴⁰. Stucke and Grunes (2016, p. 268) welcomed this new tool by stating that "*the elegance of the model is that it takes efficiencies into account, does not require the actual measurement of privacy (or quality more broadly) and has unambiguous results*". It remains to be seen whether competition authorities would use these modified tools in their assessment to deal with challenges associated with new dimensions of competition.

All in all, when it comes to mergers in two-sided online markets, focusing only on the price effects and/or monetized side of the market is prone to reach misleading conclusions. In order to assure well-founded decisions, the competition authorities should directly account for reduction in privacy on the consumer side through balancing the potential pro-competitive effects and the degradation in privacy protection by a small, but significant, non-transitory degree. The net effect of the merger on the consumer welfare would depend on how much the consumers value one side over the other. Refraining form accounting for any post-merger privacy harm may be one of the major drawbacks of the standing merger control enforcement.

4. Incorporating a more dynamic approach into merger analysis: what does it mean?

According with Ginsburg and Wright (2012, p. 1-3), dynamic competition models refer to the relationship between present competitive activities and the prediction of future market conditions such as "entry, investment, innovation, price, output, and quality"⁴⁴¹. As defined by Sidak and Teece (2009), an analysis that favors dynamic competition over static competition would place less weight on market share and concentration and more weight on assessing potential competition⁴⁴² and enterprise-level capabilities⁴⁴³. Dynamic analysis matters the most when markets are in turmoil, experiencing significant technological change⁴⁴⁴.

⁴⁴⁰ Waehrer (2016) bases its model on following assumptions; (i) quality competition, in terms of privacy protection, (ii) services are offered to consumers in two-sided online markets, (iii) revenue is generated from advertising; and (iv) merger reduces the incentive to make costly quality improvements that would make the service more attractive to the user.

⁴⁴¹ According with the authors, the term "dynamic analysis" has been used in at least two different ways: (i) to incorporating the creation of new products and business models into the static model of competition; and (ii) the relationship between present competitive activities and future market conditions.

⁴⁴² Sidak and Teece (2009) argue that competition authorities should take potential competitors and their capabilities more seriously, as new entrants almost always drive innovation in established industries.

⁴⁴³ Id. According with the authors, as capabilities transcend products, they are defined as "upstream resources" and are a better proxy for the firm's competitive position than is its downstream market share" (p. 38). In a dynamic context, a firm will have a kaleidoscope of products, yet the underlying capabilities are likely to be more stable (p. 39).

⁴⁴⁴ Favoring dynamic versus static competition Schumpeter (1942) observed: "competition from the new commodity, the new technology, the new source of supply, the new type of organization— competition which

Empirical analysis of mergers has been traditionally dominated by static microeconomic theory, holding fixed the set of incumbent firms and products in the market⁴⁴⁵⁻⁴⁴⁶. However, there are important reasons why merger analysis should further incorporate a dynamic approach. In the merger context, the static criterion to assess competition is the immediate price effects in a given market (as well as change in market shares), while a dynamic evaluation also considers the innovation process (Marshall and Parra, 2016), as well as post-merger changes in firms' incentives and behavior.

From a dynamic perspective, it should be recognized that a reduction in competition that increases R&D may harm consumers in the short-run, but may bring long-term consumer welfare in terms of new products (Marshall and Parra, 2016). Notwithstanding, it should also be recognized that, e.g., a merger that allows for near-term gains has to be weighted against eventual long-term losses. Thus, even if a transaction does not raise competitive concerns from a static perspective (i.e., price increase or higher concentration in the product market), it can harm consumer welfare in the long run.

As observed by Sidak and Teece (2009, p. 41): "[t]he question should be framed not in terms of whether product-market competition will be impaired, as that is too much of an immediate concern, but in terms of whether capabilities will be brought under unitary control, thereby possibly thwarting future variety in new product development". As recognized by the *Autoritat Catalana de la Competència* (2016), a dynamic perspective of competition it is desirable to assess to what extent the integration would boost the incumbent's data advantage, hinder rivals from access to viable alternatives, or allow for exclusionary practices, limiting innovation in the near future.

This sort of exercise is imperative in data-driven mergers for at least two reasons: (i) data can be considered as a market in itself; and (ii) a dominant position in one data-driven market can be used to gain a dominant position in a second market that is (initially) not data-driven (Prüfer and Schottmüller, 2017).

4.1 Forward-looking approach to market definition

commands a decisive cost or quality advantage and which strikes not at the margins of the profits and the output of existing firms, but at their foundations and their very lives."

^{. &}lt;sup>445</sup> According to OECD (2007, p. 22), static efficiencies are allocative and productive efficiencies which relate to a particular point in time, while dynamic efficiencies are processes that occur over time or multiple time periods and lead to lower costs, new products, or improved products, e.g., innovation and learning by doing. For more details on the meaning of static versus dynamic competition see Sidak and Teece (2009).

⁴⁴⁶ Merger analysis incorporates predictive fact-finding in terms of the likelihood of a successful entry, and the effects on prices post merger. To a less extent, competition authorities also apply a forward-looking analysis of future effects of efficiencies and innovation. See Ginsburg and Wright (2012).

Except for Twitter, platforms like Amazon, Google, and Facebook do not currently trade data with third parties, which seems to be the reason why the EC has defined the markets for, e.g., targeted ads, search engines and social networks, but not for user data⁴⁴⁷. However, "by defining a wider market for data, competition authorities and courts will be able to take a form of potential competition into consideration whereby online platforms providers also compete in a market for data" (Graef, 2015, p. 492).

As noticed by Graef (2015), a dynamic approach to market definition would be useful to evaluate the competitive situation beyond the relevant market for the current services offered to users and advertisers, and to assess competitive situations in a potential market for data used for improving the services provided on online platforms ⁴⁴⁸.

This goes in line with the opinion of the US FTC Commissioner Pamela Harbour, who suggested defining "a putative relevant product market comprising data that may be useful to advertisers and publishers who wish to engage in behavioral targeting." In the Commissioner's view, this market definition would be more realistic⁴⁴⁹.

4.2 Connected markets and the domino effect

Defining a *potential market for data needed to provide services for users and advertisers* would also be helpful to understand the incentives of data-driven platforms to leverage market power to an adjacent or "connected" market, and repeatedly in other markets ("domino effect"). As defined by Prüfer and Schottmüller (2017), connected markets are "situation where user information gained in market A is a valuable input to improve one's perceived product quality in market B". Moreover, "firm 1 will enter market B when it has become sufficiently dominant in market A"⁴⁵⁰.

In the discussed model, entering and dominating a connected market will be possible when firm 1: (i) develops a service or product that makes good use of user information gained

⁴⁴⁷ For instance, see *Facebook/WhatsApp* (2014) and *Google/DoubleClick* (2008).

⁴⁴⁸ For more details on the possibility of defining a "potential market for data needed to provide services to users and advertisers" see Graef (2015, p. 493).

⁴⁴⁹ Dissenting Statement of Commissioner Pamela Jones Harbour, *Google/DoubleClick*, FTC File No. 071-0170,
20 Dec. 2007, p. 9, http://www.ftc.gov/sites/default/files/documents/ public_statements/statement-matter-google/doubleclick/071220harbour_0.pdf.

⁴⁵⁰ "[W]e applied the model and exemplified the domino effect by showing that Google's strategy to invest in many apparently unrelated markets can be rationalized by our model: these markets are either already connected (by user information driving indirect network effects in each of them) or the firm is trying to identify business models where user information from existing markets can serve as a valuable input into traditional markets." (Prüfer and Schottmüller, 2017, p. 20-31).

in one's original market; and (ii) possess a lot of relevant user information in its home market (Prüfer and Schottmüller, 2017, p. 17). This theory largely explain Google's successful business model of acquiring and entering markets far from its core business (e.g., driverless cars, "smart home" appliances, and healthcare⁴⁵¹), but that can all benefit from access to a common pool of user information.

After the antitrust dismantle of conglomerates in the 1980s, intervention in conglomerate markets has been exceptionally rare (OECD, 2007). But the rational that conglomerate mergers do not change the incentives of the merged firm to change its behavior (Bork, 1978) does not seem to hold in the context of data-driven markets: what may looks like a messy conglomerate on the surface, can have a lot more synergies underneath. Due to data-driven indirect network effects, a conglomerate merger can allow higher concentration in the potential market for user information and the possibility of tipping in correlated markets.

4.3 Do we need new tools?

We do not claim that competition authorities should always favor dynamic competition efficiencies over static competition⁴⁵², or that they must try to accurately predict technological evolution⁴⁵³. Competition authorities should not rely completely on the reasoning presented by the merging parties either. Which is certain is that by disregarding dynamic effects a competition authority may forego potential consumer benefits or harm from these effects (OECD 2007, p. 226), especially in markets that are rapidly evolving and in areas where confidence in predictive fact-finding is supported by sound economic theory and empirical evidence (Ginsburg and Wright, 2012).

As framed by Commissioner Margaret Vestager (2016), to tackle those issues, "we don't need a whole new competition rulebook for the big data world. (...) what we do need is to pay close attention to these markets and to take action when it's necessary".

⁴⁵¹ See "Google transforms into new Alphabet tech conglomerate", Financial Times, by Richard Waters and Eric Platt, August 11, 2005, https://www.ft.com/content/9a291bf8-3fa2-11e5-b98b-87c7270955cf.

⁴⁵² According with Sidak and Teece (2009), dynamic competition can be viewed as Schumpeterian competition, while static competition can be viewed as neoclassical. In any case, economic theory does not allow us to draw a clear-cut causal relationship between structure, competition and innovation. While most authors conclude that competition is more likely to foster innovation (Arrow, 1962), others argue that, if the profit gap is increasing, mergers increase industry's incentive to innovate (Marshall and Parra, 2016).

^{. &}lt;sup>453</sup> As observed by OECD (2007, p.19), unfortunately, "dynamic efficiencies also tend to be extremely difficult to identify, substantiate, and measure, making their incorporation in merger analysis problematic. (...) They include the uncertainty inherent in innovative activity regarding its cost, timing, and likelihood and extent of commercial success, difficulties in measuring innovation itself, the problem of how to conceptually transform innovation into some measure of welfare, the challenge of comparing near-term price effects with long-term non-price effects such as quality improvements or new products, (...) and informational asymmetry between the merging parties and the enforcement agencies."

Despite the institutional difficulties and personal limitations, competition authorities can overcome the challenges of dynamic analysis by: (i) compelling customer, suppliers, and rivals to identify potential competitors and possible technological developments; (ii) conducting industry-specific studies on innovation; (iii) hiring industry experts to become informed about unfolding innovation; and (iv) conduct consumer surveys to determine what kind of product development consumers would (not) value. Those measures can help authorities to better construct likely hypothetical scenarios about how a proposed data-driven merger can change the future market conditions.

More specifically, in **Section 5** the merger between Facebook and WhatsApp is revisited as an exercise to test what could have been done differently if future market conditions would have been taken into account.

5. Could a dynamic analysis have changed the outcome of Facebook/WhatsApp?

As reported by OECD (2016), with WhatsApp owning the leading messaging platform and Facebook offering the most widely used social network, the merger between the two companies has been a focal point in the debate about big data, competition and privacy.

On October 3, 2014, two months after its notification, the EC cleared the merger between Facebook and WhatsApp. At that time, Facebook had 1.3 billion users, while WhatsApp had around 600 million users⁴⁵⁴. The EC assessed the impact of the transaction on three services: (i) consumer communications; (ii) social networking; and (iii) online (non-search) advertising. In sum, the EC concluded that the parties were distant competitors in markets (i) and (ii), and that consumers and advertisers would continue to have a wide choice of alternatives in, respectively, markets (i) and (iii) post-merger.

The transaction was approved even considering the possibility of automated user matching – which was denied at the time of the notification⁴⁵⁵ –, as a large amount of internet user data valuable for advertising would continue to exist. However, in August 2016, WhatsApp announced that it would start disclosing the phone number and analytics data of its users to

⁴⁵⁴ According to para. 128 and FN 76 f the EC's Decision.

⁴⁵⁵ At the time of the acquisition, Facebook said that it was not technically possible to match WhatsApp users' ID with Facebook accounts because most people did not load the phone number used to register on WhatsApp onto their Facebook profile. Subsequently, in August 2016, WhatsApp announced, among other updates on its terms of service and privacy policy, the possibility of linking WhatsApp user phone numbers with Facebook user identities. See, EC Press release on May 18, 2017, http://europa.eu/rapid/press-release_IP-17-1369_en.htm.

Facebook⁴⁵⁶. In May 18, 2017, EC fined Facebook \in 110 million for providing misleading information regarding the alleged technical impossibility of matching users of both platforms. Finally, in the EC's view, any privacy-related concerns flowing from the increased concentration of data within the control of Facebook as a result of the merger do not fall within the scope of the EU competition law rules but within the scope of the EU data protection rules⁴⁵⁷.

We believe that the adoption of a more dynamic oriented approach could have included the following subjects in this merger analysis: (i) a forward-looking relevant market definition; (ii) the elimination of potential competition; (iii) connected markets and domino effect; and (iv) quality degradation in terms of lower privacy protection.

5.1 A dynamic market for data

Even without defining and analyzing a market for the provision of data, the EC concluded that post-merger, "*there will continue to be a large amount of Internet user data that are valuable for advertising purposes and that are not within Facebook's exclusive control*"⁴⁵⁸. The EC referred to data collection across the web in general, without differentiating between different types of advertising, considering Google, Apple, Amazon, eBay, Microsoft, AOL, Yahoo, Twitter, IAC, LinkedIn, Adobe and Yelp as market participants that collect user data alongside Facebook⁴⁵⁹. As saw in **Section 2.2.**, the first caveat is that not all those firms collect the kind of data that can compete with the granular, up-to-date user data collected by Facebook.

From a static perspective, were the market to be defined as "non-search advertising on social network", as initially ventilated by the EC Decision⁴⁶⁰, it could be demonstrated that the data collected by the platforms mentioned above are not substitutable from the demand or producer side⁴⁶¹. This is because, as seen in **Section 2.2**., Facebook collects data on social

⁴⁵⁶ According to WhatsApp, data-sharing will allow Facebook to use a person's phone number to improve other Facebook-operated services, such as making new Facebook friend suggestions, or better-tailored advertising. See Isaac and Scott (August, 2016) and NYT (2016).

⁴⁵⁷ See para. 164, p. 29, EC Decision.

⁴⁵⁸ See para. 189, EC Decision.

⁴⁵⁹ See para. 188-189, EC Decision.

⁴⁶⁰ See para. 76 an 77, EC Decision: (76) (...) search and non-search ads are not substitutable as they serve different purposes (for search ads, mainly generating direct user traffic to the merchant's website, while, for non-search ads, mainly building brand awareness) and, as a result, most advertisers would not be likely to switch from one type to another in the event of a 5-10% price increase. (77) (...) A number of respondents considered that other forms of non-search advertising are not as effective as advertising on social networking websites and notably on Facebook, due to Facebook's large and highly engaged audience and its ad targeting opportunities.

⁴⁶¹ Twitter data seems like the only good substitute for Facebook data - although the inverse does not seem to hold. See the arguments made by John David Rich in *PeopleBrowsr v. Twitter* (2012) affirming that tweets are *"contemporaneous reports on users' experience that provide unique feedback regarding consumers' reactions to*

graph, interactions, and profile information, while Apple, Amazon and eBay collect data mainly on purchase behavior, and Google, Yahoo/Bing on search queries and clicked links. Data collected by search engines can be use to provide online search advertising but is hardly be used to provide social networking advertising.

From a dynamic perspective, if the EC had gone beyond the current services being offered by the merging parties, and realized that users' data, as an upstream resource, is a better thermometer of competitiveness than downstream market shares, a relevant market for data could have been defined.

Firstly, following the approach suggested in **Section 4.1.**, if we identify the market as "data needed to provide non-search advertising and relevant services in social network", Facebook/WhatsApp merger would render Facebook dominance even more indisputable for both sides of the platform. Unlike the EC, the author does not see how "Google+, LinkedIn, MySpace, Pinterest and InterNations"⁴⁶² could impose a competitive constraint to Facebook's near 2 billion users platform, as none of them have similar scale or network effects and, in the case of LinkedIn, it is a professional, and not a social network as Facebook. In this scenario, the merger analysis would likely conclude that Facebook is dominant in the market for advertising on social networks and, thus, integrating with WhatsApp's data could have enhanced Facebook's market power.

Secondly, it is worthy to note that even if the market was more widely defined as "data needed to provide non-search advertising", the merger would reinforce a duopoly and lead to weaker contestability as only Google could pose a competitive constraint in this market⁴⁶³.

5.2 Elimination of potential competition

As pointed out by many critics⁴⁶⁴, the reason why Facebook was paying nearly \$22 billion to buy a firm with modest revenues⁴⁶⁵ and less than 60 employees, lies in the fact that WhatsApp, by scanning millions address books, had built an alternative "social graph", the network of connections between friends, which is indeed Facebook's most valuable asset.

product and brands" and that the possibility of respond and retweet "provides unique insight about which members of communities are influential).

⁴⁶² See para. 62, EC Decision.

⁴⁶³ This report also disagrees with the EC Decision that "Yahoo!, MSN and local providers" would represent a "sufficient number of alternative" to compete against Facebook ad targeting opportunities and high return on investments. See para. 177, EC Decision.

⁴⁶⁴ The Economist (May, 2017), EDPS Opinion 8/2016, among others.

⁴⁶⁵ After transactions like Facebook/WhatsApp, authorities have incorporated an additional threshold based on the value of the transaction to the current turnover threshold. See OECD (2016, p. 20) and *Monopolkommission* (2015).

WhatsApp functionalities were becoming closer to offer a broader digital social experience, as provided by Facebook⁴⁶⁶. Therefore, Facebook was eliminating a nascent threat⁴⁶⁷ in the social network services, and the merger would rend Facebook's dominant position in social network even less contestable in the future.

Also, in the market for online advertising services, the merger was excluding the possibility of WhatsApp to serve non-search ads on its platform as an independent competitor. As observed in the dissenting statement in the FTC approval of Google/DoubleClick, also in Facebook/WhatsApp merger with respect to non-search targeted ads, the indirect network effects may not have been taken into account and the barriers to entry raised by the merger would mean that the advertisers would not have any alternative but to resort to the merged entity⁴⁶⁸.

5.3 Tipping a connected market

As the EC frames the Facebook/WhatsApp merger, there was no concentration problem in the market for target advertising as automated matching Facebook users' ID with WhatsApp users' ID was not technically possible, and even if merging data was possible, WhatsApp "limited" user information could not add much value to Facebook's data hoard⁴⁶⁹.

First, WhatsApp's data (i.e., user name, picture, status message, phone number, agenda, etc.) could be used to improve Facebook relevance of service and future advertising purposes in the future. As data collected via mobile data analytics is more personal, geo-located, and can be cross-referenced with call behavior, it could help Facebook to improve some of its functionalities, like suggesting friends, as it is actually doing⁴⁷⁰. Merging data could also be relevant for launching new AI services in the future.

Secondly, from Facebook's perspective, more crucial than to improve its data was to start serving ads in other markets. As pointed out by Fiegerman (2017), before the transaction, Facebook was hitting its maximum capacity for how many ads it could serve in its social network platform (included Instagram). Likely, the merger would expand Facebook's presence

⁴⁶⁶ As argued by many third parties, allowing for video calls, content exchange, creation of big groups, desktop access, etc., it seems that WhatsApp was already a provider of social networking services and should have been considered as a competitors of Facebook. See para 144, EC Decision.

⁴⁶⁷ The Economist (May, 2017), p. 9;

⁴⁶⁸ Dissenting Statement of Commissioner Pamela Jones Harbour in *Google/DoubleClick*.

⁴⁶⁹ See paras 71, 180-188, EC Decision.

⁴⁷⁰ See <u>http://www.nytimes.com/2016/08/26/technology/relaxing-privacy-vow-whatsapp-to-share-some-data-with-facebook.html? r=1.</u>

in the mobile target advertising, and allow it to serve ads not only in its own communication app (i.e., Messenger⁴⁷¹), but also in WhatsApp's⁴⁷².

Thus, as suggested by the theory (Prüfer and Schottmüller, 2017, p. 16-17) provided in **Section 4.2.**, Facebook was looking for a connected market to develop a service or product (i.e., chatbots) that makes good use of user information gained in Facebook's original market. Thus, evidence and economic theory indicates that Facebook's main reason to acquire WhatsApp seems to leverage its granular user data already collected on Facebook in order to start serving targeted ads in WhatsApp, a connected market (initially) not data-driven. As saw in the theory, due to indirect network effects and the decreasing marginal cost of innovating, Facebook can cause the consumer communication app market to tip in favor of WhatsApp.

Thus, post-merger, services like Telegram would no longer be a relevant competitive constraint to WhatsApp⁴⁷³, which will be able to use its collective data with Facebook to innovate much cheaper and capture full demand in the consumer communication apps market (**Section 2.4**.).

5.4 Privacy and quality degradation

Facebook/WhatsApp represented a merger between two consumer communication apps that offered consumers different levels of privacy protection. While WhatsApp kept messages encrypted, Facebook's Messenger extensively collected users' data for advertising activities. As noted by Das and Kramer (2013), *even when users delete a message, it continues to be stored on Facebook's servers*. This privacy differentiation precisely explained why many users signed up for WhatsApp instead of Messenger (NYT, 2016).

As the services offered different privacy policies, the EC considered that Facebook's Messenger and WhatsApp were not close competitors, and that consumers would continue to have a wide choice of alternative consumer communications apps after the transaction. Thus, the EC, as well as the FTC, cleared the merger without further assessment on the possibility of

⁴⁷¹ Spring, Facebook's mobile shopping start-up, is an interface that allow users to begin a personal conversation with a "chatbot" that will show them a smattering of thing they may like. See Isaac (April, 2016).

⁴⁷² WhatsApp blog announced on August 25, 2016, that, although users phone number and encrypted messages stay private, it was changing its terms and privacy policy to allow it to "coordinate more with Facebook", and test new ways for its users to "communicate with businesses" in the months ahead. See https://blog.whatsapp.com/10000627/Looking-ahead-for-WhatsApp.

⁴⁷³ This report also disagrees with the EC's point of view that LINE, WeChat, iMessage, Snapchat could represent a significative competitive constraint to WhatsApp 1 billion users platform, as they have much less users and user engagement, and Skype, Viber and Hangout are not mainly used for content and messages exchange, but for video calls.

future privacy degradation, under the condition that WhatsApp obtain users' consent before changing privacy policies.

Two years after the merger was approved, in contradiction to the information submitted to the EC by the merging parties, WhatsApp announced that it would start sharing users' information with Facebook. This report argues that a more forward-looking inquiry by the EC, as well as a more transparent conduct by the merging parties, could have avoided this tight spot for the authority and for consumers, even if the EC did not considered WhatsApp data relevant for Facebook activities.

Specifically, the parties were close competitors in the market for consumer communication apps, regardless of the level of privacy offered (Waehrer, 2016). Secondly, privacy was precisely the most differentiating factor of competition between those services. The purchase of WhatsApp's privacy-friendly model is likely to deprive consumers of meaningful privacy choices (OECD, 2016, p. 18), and preventing WhatsApp from breaking out a "dysfunctional equilibrium" created by Facebook (Stucke and Grunes, 2016). Thus, the EC should have scrutinized whether Facebook lessened competition by eliminating a possible maverick firm (i.e., WhatsApp) offering greater privacy protection.

Accordingly, a more dynamic approach would have urged the EC to analyze whether consumers would be harmed by the privacy degradation if Facebook match WhatsApp users data⁴⁷⁴, even if there was no breach of privacy laws. In particular, the EC could have balanced the gain to the advertisers against the loss in quality to consumers (OECD, 2016, p. 16). Also, the EC could have analyze to what extent users would be able to detect quality degradation and switch to alternative apps offering privacy protection given the network effects present in the case, as consumers are in an even worse condition to assess the value of their data, and often have no option but to accept the terms imposed on them, as well as intrusive advertising and even behavioral discrimination (The Economist, September, 2016).

Notwithstanding its challenges, a "harder-to-measure but equally important" non-price parameters of competition such as privacy protection (Stucke and Grunes, 2016), could yield a more comprehensive analysis than the one offered by price centric tools. All in all, an analysis that considered the impact of privacy reduction on consumers' choice would likely prevent the

⁴⁷⁴ For instance, the Electronic Privacy Information Center, in its comments to FTC states that "*it was clear that the practical consequence of the merger would be to reduce the privacy protections for consumers and expose individuals to enhanced tracking and profiling. The failure of the Federal Trade Commission to take this into account during merger review is one of the main reasons consumer privacy in the United States has diminished significantly over the last 15 years*" In re: Remedy Study "Assessment of the FTC's Prior Actions on Merger Review and Consumer Privacy" March 17, 2015, https://epic.org/privacy/internet/ftc/Merger-Remedy-3-17.pdf.

acquisition by a dominant platform of companies providing services with a greater extent of privacy protection (OECD, 2016, p. 19).

6. Conclusion

While big data may boost innovation and connectivity, it can also entrench market power and compromise privacy. This tradeoff can be better addressed by a deeper integration of dynamics effects into competition analysis. In this sense, competition authorities could evaluate the market for data itself, as well as better account for potential competition and future market conditions. As suggested with the re-evaluation of Facebook/WhatsApp, a more dynamic analysis, however challenging and imperfectly, is better equipped to account for datadriven mergers' effects on consumer welfare in the long-run (Sidak and Teece, 2009; and Buttarelli, 2016).

In particular, regarding the analysis of data-driven mergers, competition authorities could take into account that: (i) data can yield an unreplicable advantage and entrench market power, as information about users' last minute behavior is not easily or readily available; (ii) dominant platforms nowcast not only consumer's behavior, but also the development of rivals' business models, precluding nascent threats from a chance to displace incumbents; and (iii) data-driven indirect network effects can cause the product market or a correlated market to tip much more easily, hindering entry.

With respect to privacy-related concerns, competition authorities could recognize that: (i) privacy standards are a growing concern of consumers, a cognizable parameter of competition, and a differentiation factor for competition; (ii) where services are offered for free, consumers may feel forced to accept lower standards of quality (i.e., privacy); (iii) the lack of competition over privacy implies market failure (CMA, 2015), and; (iv) competition authorities should intervene when privacy is a relevant dimension of competition and consumers' choices are overall restrained.

7. References

ACQUISTI, Alessandro et al. (2016), The Economics of Privacy, Journal of Economic Literature, 54(2), 442–492.

ARGENTON, C. and PRÜFER, Jens (2012), Search Engine Competition with Network Externalities, Journal of Competition Law and Economics, vol. 73.

ARROW, Kenneth J. (1962), Economic Welfare and the Allocation of Resources for Invention, in The Rate And Direction Of Inventive Activity: Economic and Social Factors, Richard Nelson ed.

ASAY, Matt and O'REILLY, Tim (2010), Whole Web is the OS of the future, CNET.

AUTORITAT CATALANA DE LA COMPETÈNCIA (November, 2016), The Data-Driven Economy. Challenges For Competition.

BAYE, Michael R. et al. (2008), "Economics at the FTC", Review of Industrial Organization (33:3), p. 211-230.

BORK, Robert H. (1978) The Antitrust Paradox: A Policy at War with Itself, New York: Basic Books.

BUTTARELLI, Giovanni (September 29, 2016), Big data: individual rights and smart enforcement, speech at the EDPS-BEUC Joint Conference European Commission, Berlaymont, Brussels.

COMPETITION AND MARKETS AUTHORITY (May, 2015), The commercial use of consumer data: report on the CMA's call for information.

CORNES, R., and SANDLER, T. (1968). The theory of externalities, public goods, and club goods. Cambridge University Press.

DAYEN, David (April 4, 2017), This budding movement wants to smash monopolies, The Nation.

DAYEN, David (March, 2016), The Most Important 2016 Issue You Don't Know About, New Republic.

DAS, Sauvik and KRAMER, Adam (2013), Self-Censorship on Facebook, Association for the Advancement of Artificial Intelligence, 2013. Das & Kramer.

EUROPEAN UNION (April 27, 2016), General Data Protection Regulation 2016/679, European Parliament and of the European Council.

EDPS (September 23, 2016) Opinion on coherent enforcement of fundamental rights in the age of big data 8/2016, European Data Protection Supervisor.

EVANS David S., (2011), The Antitrust Economics of Free, Competition Policy International (2011), May 17.

EZRACHI, Ariel, and STUCKE, Maurice E. (2016), Virtual Competition: the promise and perils of the algorithm-driven economy, Cambridge, Massachusetts: Harvard University Press.

EZRACHI, Ariel, and STUCKE, Maurice. E. (2015), The Curious Case of Competition and Quality, Journal of Antitrust Enforcement, 3:2, 1-31.

FARRELL, J. (2012), Journal on Telecommunications and High Technology Law, vol 10:2, p. 251-264.

FIEGERMAN, Seth (May 3, 2017), Facebook tops 1.9 billion monthly users, CNN Tech. FIRICAN, George (February 8, 2017), The 10 Vs of Big Data, UpSide.

GAL Michal S. & RUBINFELD Daniel L. (2016), The Hidden Costs of Free Goods: Implications for Antitrust Enforcement, Antitrust Law Journal Vol. 80 Iss.3 p.521-562

GINSBURG, Douglas H., and WRIGHT, Joshua D. (2012), Dynamic Analysis and the Limits of Antitrust Institutions, Antitrust Law Journal, vol. 78, No. 1, 2.

GRAEF, Inge (2015), Market Definition and Market Power in Data: the case of online platforms, World Competition 38, No. 4, p. 473-506.

HARTMAN R., TEECE D., MITCHELL W. & JORDE T., (1993) "Assessing Market Power in Regimes of Rapid Technological Change," 2 Industrial and Corporate Change.

HESSE, Renata (September 20, 2016), And Never the Twain Shall Meet? Connecting Popular and Professional Visions for Antitrust Enforcement, Opening Remarks at 2016 Global Antitrust Enforcement Symposium Washington, DC.

ISAAC, Mike, and Scott, Mark (August 25, 2016) Relaxing Privacy Vow, WhatsApp will share data with Facebook, The New York Times.

ISAAC, Mike, and SCOTT, Mark (August 25, 2016), Relaxing Privacy Vow, WhatsApp will share some data with Facebook, The New York Times.

ISAAC, Mike (April 12, 2016), Facebook bets on Bots for its Messenger App, The New York Times.

KHAN, Lina M. (2017), Amazon's Antitrust Paradox, The Yale Law Journal, vol. 126, No. 3

KATZ, Michael L. and SHAPIRO, Carl (1985), Network Externalities, Competition, and Compatibility, American Economic Review, vol. 75, issue 3, p. 424-40.

KIMMEL L & KESTENBAUM J (2014), What's Up with WhatsApp? A Transatlantic View on Privacy and Merger Enforcement in Digital Markets, 29 Antitrust 48.

KENNEDY, Joe, (2017), The Myth of Data Monopoly: Why Antitrust Concerns About Data Are Overblown, MARCH 2017, Information Technology & Innovation Foundation.

LANEY, Doug (2001), 3D data management: Controlling data volume, variety and velocity, MetaGroup research publication.

LANIER, Jaron (2014). Who Own the Future? Published by Simon & Schuster.

LAMADRID, Alfonso, and VILLIERS, Sam (January 2017), Big Data, Privacy and Competition Law: Do Competition Authorities know how to do it? Competition Policy International, Antitrust Chronicle.

MANNE A. Geoffrey & SPERRY R. Ben, (May 2015), The Problems and Perils of Bootstrapping Privacy and Data into an Antitrust Framework, CPI Antitrust Chronicle.

MARSHALL, Guilhermo, and PARRA, Álvaro (2016), Mergers in Innovative Industries: The Role of Product Market Competition.

MONOPOLKOMMISSION (2015), "Competition Policy: The Challenge of Digital Markets", Special Report No. 68 by the Monopolies Commission pursuant to the section 44(1)4 of the Act against Restrains on Competition.

NEWMAN, John M. (2014), Antitrust in Zero-Price Markets: Foundations, SSRN Working Paper, 32-33.

OCELLO, E., C. SJÖDIN and A. SUBOČS (2015), "What's Up with Merger Control in the Digital Sector? Lessons from the Facebook/WhatsApp EU Merger Case", Competition Merger Brief No. 1/2015.

OECD (May, 2007), Dynamic Efficiencies in Merger Analysis, Policy Roundtables, Competition Committee.

OECD (November 2016), Big Data: Bringing competition policy to the Digital Era, Competition Committee.

OECD (October, 2014), Data-driven Innovation for Growth and Well-being, Interim Synthesis Report.

OECD (2013), The Role and Measurement of Quality in Competition Analysis, Competition Committee.

OHLHAUSEN Maureen K. & OKULIAR, Alexander P. Competition, Consumer Protection, And The Right [Approach] To Privacy.

PASQUALE, Frank (2013), Privacy, Antitrust, and Power, Geo. Mason L. Rev, Vol 20:4. PRÜFER, Jens, and SCHOTTMÜLLER, Christoph (2017), Competing with Big Data, Tilburg Law School Legal Studies Research Paper Series No. 06/2017.

SCHMIDT, Eric (January 20, 2015), 'Why Google Works', Huffington Post.

SCHUMPETER, Joseph A. (1942), Capitalism, Socialism and Democracy, New York: Harper & Row.

SHAPIRO, Carl and VARIAN, Hal R. (1999), Information Rules: a strategic guide to the network economy, Harvard Business School Press, vol. 3.

SOKOL D. Daniel & COMERFORD Roisin, Does Antitrust Have A Role to Play in Regulating Big Data?

SIDAK, Gregory, and TEECE, David (2009), Dynamic Competition in Antitrust Law, Journal of Competition Law & Economics, Vol. 5, No. 4, pp. 581–631.

STUCKE, Maurice E., and GRUNES, Allen P. (2017), Data-opolies, Legal Studies Research Paper Series, No. 316.

STUCKE, Maurice E. and GRUNES, Allen P. (2016) Big Data and Competition Policy, Oxford University Press.

STUCKE, Maurice E. and GRUNES, Allen P. (May 31, 2015), Debunking the Myths Over Big Data and Antitrust. CPI Antitrust Chronicle.

STUCKE, Maurice E. and GRUNES, Allen P. (April 2015), No Mistake About It: The Important Role of Antitrust in the Era of Big Data, The Antitrust Source.

TAPLIN, Jonathan (2017), Move Fast and Break Things: How Facebook, Google and Amazon have cornered culture and what it means for all of us, published by Mcmillan.

THE ECONOMIST (March 25, 2017), Levelling the paying field – New European rules herald a welcome challenge to incumbent retail banks, p. 14 and p. 65.

THE ECONOMIST (May 6, 2017). Fuel of the future: data is giving rise to a new economy & Price-bots can collude against consumers.

THE ECONOMIST (September 17, 2016), The superstar company: A giant problem. THOMPSON, Derek (October 2016), America's Monopoly Problem: how big business jammed the wheels of innovation, The Atlantic.

TUCKER, Darren S. and WELLFORD, Hill B. (December 1, 2014), Big Mistakes Regarding Big. Data, Antitrust Source.

WAEHRER, Keith. (2016) Online services and the analysis of competitive merger effects in privacy protections and other quality dimensions. Bates White, LLC.

Case law:

Asnef-Equifax v. Ausbanc, C-238/05, ECLI:EU:C:2006:734 Facebook/WhatsApp, COMP M.7217, EC's Decision on 03.10.2014 Google/DoubleClick, COMP/M.4731, EC's Decision on 22.07.2008 Microsoft/Skype, M COMP/M.6281, EC's Decision on 07.10.2011 Post Danmark A/S v. Konkurrencerådet, C-209/10, Judgment of the ECJ on 27.03.2010 PeopleBrowsr v. Twitter, Superior Court of the State of California, November 2012